Author’s response to reviews

Title: A novel real-time IVUS double-lumen microcatheter for recanalization of chronic total occlusion: A case report

Authors:

Yanzhuo Ma (muyeqiu2008@163.com)
MD Peng (1360247947@qq.com)
MD Wang (qwqelihaoliang@163.com)
Leisheng Ru (297400148@qq.com)

Version: 1 Date: 25 Jul 2019

Author’s response to reviews:

Thank you for your careful evaluation of our manuscript and many constructive suggestions.

Comment: Well-presented article, but will need major correction for English grammar and sentence rewriting in accepted scientific literature.

Response: We have carefully revised the manuscript and corrected the typographical and grammatical errors. As suggested by the reviewer, our manuscript has been proofread by a scientific editor in the revised manuscript.

Comment: Some words also need to be revised like Ostium instead of ostial.

Response: We completely agree with the reviewer. This word has been revised appropriately.

Comment: Also you need to mention the manufacturer of this device. If this is not commercially made, then you have insert the patent numbers and the company that helped put it together.

Response: This device is not commercially made. We have inserted the patent numbers and the name of the company that helped put it together, as suggested by the reviewer (page 3, line 15).